Author's response to reviews

Title: Efficacy and safety of ondansetron in preventing postanesthesia shivering: a meta-analysis of randomized controlled trials

Authors:

Hong-Tao Tie (hongtaotie@163.com)
Guang-Zhu Su (suguangzhu@163.com)
Hao-Wei Yuan (yuanhw568@163.com)
Jun-Huan Mou (junhuanmou@126.com)
Kun He (kun_he@126.com)
Shao-Rong Liang (shaorong_liang@126.com)

Version:3 Date: 12 February 2014

Author's response to reviews: see over
Dear Prof. Marielette Costoy

I appreciate your instructive comments to improve the manuscript (ID: 1682543311019849). We have addressed your concern. We assure that our manuscript conforms to the PRISMA guidelines and have uploaded the copy of the PRISMA checklist of our manuscript. Now, we determine to resubmit it to *BMC Anesthesiology* for publication.

We expect this paper can be published in the *BMC Anesthesiology*.

Thank you very much!

Sincerely yours

Hong-tao Tie